

JGR Solid Earth

RESEARCH ARTICLE

10.1029/2024JB029004

Key Points:

- Non-parametric remanent magnetization curves unmixing provides physical, though ambiguous, components
- Non-parametric Preisach maps unmixing resolves this ambiguity by allowing the identification of mineral phases and domain states
- In loess-paleosol deposits the pedogenic and eolian components are respectively independent of pedogenesis intensity and dust source areas

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

C. Necula and I. Lascu, cristian.necula@unibuc.ro; lascui@si.edu

Citation:

Necula, C., Lascu, I., Panaiotu, C., & Gheorghe, D. (2024). Resolving the interpretation of magnetic coercivity components from backfield isothermal remanence curves using unmixing of nonlinear preisach maps: Application to lossspaleosol sequences. *Journal of Geophysical Research: Solid Earth*, *129*, e2024JB029004. https://doi.org/10.1029/ 2024JB029004

Received 27 FEB 2024 Accepted 5 MAY 2024

Author Contributions:

Conceptualization: Cristian Necula, Ioan Lascu Data curation: Cristian Necula Formal analysis: Cristian Necula Funding acquisition: Ioan Lascu, Cristian Panaiotu Investigation: Cristian Necula, Ioan Lascu, Cristian Panaiotu, Daniela Gheorghe Methodology: Cristian Necula, Ioan Lascu, Cristian Panaiotu, Daniela Gheorghe Project administration: Cristian Panaiotu

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Resolving the Interpretation of Magnetic Coercivity Components From Backfield Isothermal Remanence Curves Using Unmixing of Non-Linear Preisach Maps: Application to Loess-Paleosol Sequences

Cristian Necula¹, Ioan Lascu², Cristian Panaiotu¹, and Daniela Gheorghe³

¹Faculty of Physics, University of Bucharest, Măgurele, Romania, ²Smithsonian Institution, National Museum of Natural History, Washington, DC, USA, ³Faculty of Geology and Geophysics, University of Bucharest, Bucharest, Romania

Abstract Unmixing of remanent magnetization curves, either isothermal remanent magnetization (IRM) or backfield IRM, is widely used in rock magnetic and environmental magnetic studies to discriminate between magnetic coercivity components of different origins. However, the wide range of physical properties of natural magnetic particles gives rise to an ambiguous interpretation of these components. To reduce this ambiguity and provide a straightforward interpretation of coercivity components in terms of domain state, interactions, and constituent magnetic phases, we combined backfield IRM unmixing with unmixing of nonlinear Preisach maps for two typical mid-latitude northern hemisphere loess-paleosol sequences. Both backfield IRM and nonlinear Preisach maps unmixing are based on the same non-parametric algorithm, and provide similar endmembers (EMs) in the two sections studied. The first EM (EM1) has a low median coercivity (~21 mT) and is a noninteracting single domain (SD) magnetite/maghemite of pedogenic origin. The second EM (EM2) has a moderate median coercivity (~60 mT) and is a mixture of pseudo-single domain/multidomain, SD magnetite/ maghemite and non-interacting SD hematite, all of eolian origin. The same EM1 found in both sections suggests that this component's grain size and coercivity are independent of pedogenesis intensity. The same EM2 indicates that a similar magnetic population is being transported and deposited, irrespective of the dust source area and loess granulometry. The approach outlined here provides strong evidence that non-parametric backfield IRM unmixing isolates physically realistic EMs. Unmixing nonlinear Preisach maps elucidates these EMs in terms of domain states and their constituent magnetic phases.

Plain Language Summary Magnetic minerals occurring in natural materials carry crucial information about environmental processes. However, since a given material typically contains several magnetic components, the environmental signals it encodes are convoluted. Due to different physical properties of natural magnetic particles, recovering information about each individual process can be a challenging task. One popular method used to differentiate magnetic populations with different origins is numerical unmixing of their coercivity distributions because they are sensitive to characteristics of individual magnetic populations. Coercivity distributions represent the first derivatives of IRM acquisition curves, which are built by exposing a sample to a series of magnetic fields in the laboratory. Generally, there is ambiguity in the interpretation of coercivity components. Therefore, we devised an unmixing method using data sets called non-linear Preisach maps, built from a sequence of IRM curves that are processed into two-dimensional diagrams. These data sets give us additional information to that gleaned from coercivity distributions, such as magnetic domain states and particle interaction signatures. We tested this approach on two loess-paleosol sections located in Southeastern Europe, because they developed in different climatic conditions, and had different dust source areas.

1. Introduction

Unmixing of remanent magnetization curves is extensively used to discriminate between magnetic populations with different origins and contributions to bulk magnetic mineral properties (Heslop, 2015 and references therein). It has become an essential tool for gleaning geologic and paleoenvironmental information, with wide-ranging applications in paleomagnetism (e.g., Chen et al., 2017; Roberts et al., 2012; Zhang et al., 2020), quantitative environmental magnetism (e.g., Abrajevitch et al., 2009, 2013, 2015; Hu et al., 2020; Hyland et al., 2015; Just et al., 2012; Li et al., 2020), archeomagnetism (e.g., Jordanova et al., 2020), soil and loess-paleosol magnetism studies (Geiss et al., 2008; Hu et al., 2013, 2015, 2021; Hyodo et al., 2020; Jordanova &





Resources: Cristian Necula, Ioan Lascu, Cristian Panaiotu, Daniela Gheorghe Software: Cristian Necula Supervision: Cristian Necula, Ioan Lascu Visualization: Cristian Panaiotu Writing – original draft: Cristian Necula, Ioan Lascu Writing – review & editing: Cristian Necula, Ioan Lascu,

Cristian Panaiotu, Daniela Gheorghe

Jordanova, 2020; Jordanova et al., 2011; Necula et al., 2013; Nie et al., 2014; Spassov et al., 2003), atmospheric aerosol pollution (e.g., Larrasoaña et al., 2021; Sagnotti et al., 2006), etc.

Two types of approaches have been used to differentiate between various types of magnetic components. The first approach is based on a parametric method in which the remanent curve (either isothermal remanent magnetization (IRM) or backfield IRM), or its first derivative (known as the coercivity distribution), is fitted by a linear combination of individual component distributions modeled by known mathematical functions, such as lognormal (Kruiver et al., 2001; Robertson & France, 1994; Stockhausen, 1998), skewed generalized Gaussian (Egli, 2003, 2004a, 2004b), skewed normal (SN) (Maxbauer et al., 2016b), or Burr type XII (Zhao et al., 2018) distributions. The parametric method has several drawbacks. First, the unmixing results depend strongly on the shape of basis functions chosen to model the data (e.g., Egli, 2021; Li et al., 2020; Zhao et al., 2018). Second, the quality of measurements and the initial guess for the individual components highly influence the results of the IRM unmixing (e.g., He et al., 2020; Zhao et al., 2020; Zhao et al., 2020; Read adds to the ambiguity of determining individual components and their number in the fitting process (Egli, 2021; He et al., 2020). Moreover, micromagnetic simulations have shown that the IRM curves for an assemblage of non-interacting magnetic particles with a simple grain size distribution need to be fitted by multiple Gaussian functions, due to the distribution of grain elongation and the different types of switching of particles with different shapes (Bai et al., 2021).

The second approach was proposed by Heslop and Dillon (2007) and uses a dimensionality reduction nonparametric method, namely non-negative matrix factorization (NMF). This method determines the shape of the coercivity distributions and their contributions by using only the measured data. The main advantage of the non-parametric method is that it overcomes the problems associated with selecting the number of components and types of mathematical functions to fit. Several studies have used this method to identify the main processes responsible for magnetic properties of loess-paleosol deposits (Necula et al., 2013, 2015; Nie et al., 2014), soils (Hu et al., 2020), lacustrine sediments (Fabian et al., 2016; Nie et al., 2017), or marine sediments (Just et al., 2012), as well as to diagnose remagnetization in natural samples (Dekkers, 2012). The disadvantages of this method are that the endmembers (EMs) provided may not represent single components (He et al., 2020; Heslop, 2015; Just et al., 2012), and that if the overlap of the coercivity distributions is significant, the recovery of the pure EMs and their corresponding abundances may fail (He et al., 2020; Heslop, 2015).

Several studies (e.g., He et al., 2020; Hu et al., 2021; Maxbauer et al., 2017; Roberts et al., 2018) have used a complementary approach, combining results from IRM unmixing with another non-parametric unmixing method, principal component analysis of first-order reversal curves (FORCs) (Harrison et al., 2018; Lascu et al., 2015). FORC-PCA has proven to be a powerful method that can discriminate between signatures of domain states, magnetostatic interactions, or mineral phases, providing an improved understanding of magnetic particle mixtures. However, FORCs are in-field measurements, and, in addition to remanence contributions, also record transient, induced and thermal relaxation phenomena (e.g., Harrison et al., 2019; Zhao et al., 2017). Therefore, a direct link between components resulting from IRM unmixing and domain states revealed by FORC-PCA unmixing is difficult to establish (e.g., Egli, 2021; Zhao et al., 2017).

Here we explore whether IRM coercivity components reflect one or multiple domain states by using a nonparametric unmixing method of individual backfield curves combined with a non-parametric unmixing of nonlinear Preisach maps (Church et al., 2016). Nonlinear Preisach maps are built from a sequence of backfield IRM curves, each starting from a different conditioning field, covering the entire remanence space. Therefore, integration of the Preisach function over this space is equivalent to an IRM curve measured from an initially AF demagnetized state (Church et al., 2016). In addition, no smoothing is performed on the Preisach maps, thus avoiding the artifacts associated to this type of post-measurement processing. Furthermore, nonlinear Preisach maps perform better than FORC diagrams for complex compositions where the remanent signal is masked by non-remanent contributions (Church et al., 2016).

We test this combined approach on two Romanian loess-paleosol sequences chosen to reflect different climatic conditions and different source areas, one in the Southern Wallachian Plain, and the other on the Black Sea shore (Necula et al., 2015). These sites are typical for mid-latitude northern hemisphere loess deposits, known as "Chinese loess type" (Marković et al., 2015). Magnetic properties of loess-paleosol sequences from Southeastern Europe record glacial-interglacial cyclic climatic changes on a global scale, allowing the reconstruction of past climatic conditions for the last 1 million years (Marković et al., 2015; Obreht et al., 2019; Song et al., 2018;





Figure 1. Location of the loess-paleosol sections investigated in this study: Lunca and Costineşti.

Zeeden et al., 2017). Rock magnetic studies have shown that Southeastern European loess-paleosol deposits, like their Chinese counterparts, are characterized by so-called "magnetic enhancement" in the paleosols, because of pedogenesis during interglacials (Buggle et al., 2014; Fitzsimmons et al., 2012; Jordanova & Petersen, 1999; Marković et al., 2015; Necula et al., 2013, 2015; Panaiotu et al., 2001; Zeeden et al., 2016). Loess deposits from Serbia, Bulgaria and Romania have demonstrated that this magnetic enhancement is mainly due to the presence of fine-grained, superparamagnetic and single domain (SD) ferrimagnetic minerals (magnetite and/or maghemite) produced in situ during wet and warm climate conditions. During glacials, when it is generally dry and cold, the accumulation of aeolian dust prevails, resulting in a significant input of coarse detrital magnetite in the pseudo-single domain (PSD) to multidomain (MD) range (e.g., Jordanova & Jordanova, 2020; Laag et al., 2021; Necula et al., 2015). These domain states are inferred either from bulk magnetic parameters measurements, or from IRM unmixing coercivity distributions, and typically suffer from ambiguity in interpretation (Egli, 2021). The dual IRM-nonlinear Preisach maps approach proposed here provides a straightforward interpretation of IRM unmixing EMs in terms of domain state, magnetostatic interactions, and composition, casting a new light on the populations of grains controlling the magnetic properties characteristic of the Chinese loess type.

2. Study Sites

The first loess-paleosol sequence is in the Southern Wallachian Plain near the village of Lunca, on the left bank of the Sâi River, approximately 15 km north of its confluence with the Danube River ($43^{\circ}50'56''N$, $24^{\circ}45'56''E$) (Figure 1). The section is about 34 m-thick and contains six loess layers (L1–L6), the recent soil (S0) and six intercalated paleosols (S1–S6). The section is continuous, without visible erosion or hiatuses, and is the longest loess-paleosol sequence from Romania accessible in outcrop. A chronostratigraphy of the Lunca deposit was established using optically-stimulated luminescence and a magnetic age-depth model (Constantin et al., 2015). The luminescence ages obtained on fine and coarse quartz have shown that loess L1 developed during marine isotopic stages (MIS) 2–4. The magnetic age-depth model based on the correlation of a preliminary magnetic susceptibility record with the stack of 57 globally distributed benthic δ^{18} O records (Lisiecki & Raymo, 2005) is in agreement with the luminescence ages, indicating that palaeosols S1–S4, and S6 can be correlated with the interglacials in the interval MIS 5–11, and MIS 17, respectively, while paleosol S5 spans MIS 13–15 (Constantin et al., 2015). According to this age-depth model, loess accumulation started at Lunca around 700 ka, making this sequence one of the oldest loess-paleosol sections from Romania. These findings agree with the Danube loess



stratigraphy (Marković et al., 2015) and the recent luminescence chronology of the northeastern Bulgarian loesspaleosol sequences (Balescu et al., 2019).

The second loess-paleosol sequence is located about 300 km east of Lunca, on the Black Sea shore near the village of Costineşti (Figure 1), and contains five loess layers (L1 to L5), the recent soil (S0) and five interbedded paleosols (S1 to S5). This section is about 12 m thick and covers the last ~580 ka (Constantin et al., 2013), exhibiting a sedimentation rate approximately three times lower compared to Lunca. Necula et al. (2015) showed that during the last ~580 ka the palaeosols, formed during interglacials, experienced pedogenic alteration, resulting in high amounts of superparamagnetic, SD and pseudosingle domain magnetite/maghemite grains and hematite. The loess layers, formed during glacial periods, are mainly dominated by MD and/or pseudosingle domain oxidized magnetite and some hematite, all of aeolian origin.

3. Methods

For this study we measured backfield IRM, FORCs, and grain size on the Lunca samples, and performed nonlinear Preisach measurements on both Lunca and Costineşti samples. The median grain size for Costineşti was determined from the measurements performed by Necula et al. (2015). Backfield IRM curves for Costineşti and their NMF unmixing are from Necula et al. (2015). The magnetic susceptibility data for Lunca section are from Constantin et al. (2015).

3.1. Lunca Backfield IRM Measurements and NMF Unmixing

The Lunca section was cleaned, prepared for sampling by avoiding bioturbated areas, and sampled at every 10 cm for magnetic susceptibility (Constantin et al., 2015) and granulometry measurements (a total of 336 samples). Based on magnetic susceptibility measurements we selected samples for remanence measurements at every \sim 20 cm (a total of 168 samples). Both backfield remanent moment and direct moment curves were measured as part of the same protocol using a PMC MicroMag 3900 vibrating sample magnetometer. Samples were first magnetized at 1 T to obtain an IRM and then the field was reversed and nonlinearly incremented from 0 to -1 T in 30 steps. To align IRM unmixing with Preisach map analysis we measured backfield IRM instead of IRM acquisition curves: the NLP protocol devised by Church et al. (2016) is built on a backfield protocol even if the curves are measured in forward fields. Secondly, using backfield IRM measurements allows for reproducibility of the experiment.

For backfield IRM unmixing we used the non-parametric algorithm of Heslop and Dillon (2007), which requires IRM acquisition curves as input data. To simulate IRM acquisition, we rescaled (division by a factor of 2), reversed and inverted our backfield curves (Heslop & Dillon, 2007). The algorithm also requires all IRM curves to be measured using identical measurement fields. Therefore, we used linear interpolation to anchor all the curves to the same acquisition fields. The unmixing algorithm assumes that all IRM curves can be explained by a linear mixture of a small number of EMs. The estimation of the number of EMs included in the unmixing model was based on the calculation of the coefficient of determination between the data and the model, versus the number of EMs (Heslop & Dillon, 2007). We also performed standard parametric remanence decomposition on representative soil and loess samples from Lunca (S0 and L1) using MaxUnmix (Maxbauer et al., 2016b) to facilitate the interpretation of the EMs derived from NMF analysis (He et al., 2020).

3.2. Nonlinear Preisach Maps Measurements and Unmixing

Nonlinear Preisach maps were generated from remanence curve data sets collected using the Lake Shore Cryotronics 8604 vibrating sample magnetometer at the Biogeomagnetism Lab, National Museum of Natural History, Smithsonian Institution. We prepared 24 samples, one from each loess and paleosol layer at both sites. For each sample, 50 Preisach map curves were measured at 50 steps per curve, on a nonlinear scale from 0.1 mT to 1 T, with a 1 T saturation field and an averaging time of 1 s. Nonlinear Preisach maps were calculated following the protocol of Church et al. (2016).

Unmixing of the nonlinear Preisach maps was performed using independent component analysis (ICA), as it is implemented in the HyperSpy Python library (de la Peña et al., 2011, 2021). ICA belongs to the "blind source separation" class of unsupervised analysis methods, which aim to recover EM source signals from their linear mixture without any knowledge about the mixing process or the source signals themself (Hyvärinen et al., 2001).

ICA assumes that the EMs are statistically independent, and that they have non-gaussian distributions (Hyvärinen & Oja, 2000; Hyvärinen et al., 2001). In matrix notation the mixing model can be written as

$$x = A \times s \tag{1}$$

where x is the observations matrix, A in the mixing matrix and s are the independent components. ICA model estimates the unmixing matrix

$$W = A^{-1} \tag{2}$$

such that the recovered sources

$$x = W \times x \tag{3}$$

are statistically independent. Thus a recovered independent component, *y*, consists of a linear combination of the observations contained in the matrix *x*. Mathematically can be written as

5

$$y = w^T x = \sum w_i x_i = w^T A s \tag{4}$$

where the vector w should be one of the rows of the A^{-1} matrix (Hyvärinen & Oja, 2000). According to the central limit theorem, a sum of two independent random variables is more Gaussian than the original variables. Thus y is least Gaussian when it is identical to one independent component s, and as such w can be chosen so as to maximize the nongaussianity of y (Hyvärinen & Oja, 2000). Details of the ICA unmixing algorithm and the selection of EMs can be found in Text S1 and Figure S1 in Supporting Information S1. As a pre-processing step, the data are "whitened" by transforming them into a new vector, in which the measurements are uncorrelated and their variances equal unity. The most popular method to perform the whitening is to apply singular value decomposition (SVD) directly on the data. Singular value decomposition plays two roles. First, SVD is used to determine the number of EMs, an operation ICA is not designed to perform (de la Peña et al., 2011; Hyvärinen & Oja, 2000; Hyvärinen et al., 2001). Second, once the number of EMs has been established, the first SVD factors, corresponding to the number of EMs, are used as initialization for the ICA. The latter step is done to avoid the problem of overfitting (de la Peña et al., 2011; Hyvärinen & Oja, 2000; Hyvärinen et al., 2001). To determine the EMs for the nonlinear Preisach data set, and their contributions to each sample, the paired SVD + ICA approach was applied on the Preisach curves. Prior to the application of SVD + ICA, the remanence curves were normalized to the saturation remanent magnetization. The entire procedure was performed directly on the curves rather than on the Preisach maps images (cf., Harrison et al., 2018). The SVD + ICA approach was also applied on the backfield IRM data set, to directly compare with the results obtained from the nonlinear Preisach maps unmixing. SVD + ICA was applied to the coercivity distributions with each coercivity spectrum being normalized to sum to one. Details about the unmixing procedure and selection of EMs can be found in Text S2 and Figure S2 in Supporting Information **S1**.

3.3. FORC Measurements and Unmixing

Thirteen sister samples from Lunca were used to generate FORC diagrams for the loess and paleosol layers. Firstorder reversal curves were collected using the PMC MicroMag model 3900 vibrating sample magnetometer at the Paleomagnetic Laboratory, University of Bucharest. For each sample, 150 FORCs were measured using a saturating field of 1 T, a field increment of 0.875 mT, and an averaging time of 500 ms. FORC data were processed with FORCinel 3.07 (Harrison & Feinberg, 2008), using the VARIFORC variable smoothing algorithm (Egli, 2013). FORC-PCA was conducted according to the method of Lascu et al. (2015) using the algorithms of Harrison et al. (2018).

3.4. Grain Size Measurements

Grain-size analyses of 336 samples from Lunca collected at 10 cm sampling interval were performed using a Horiba laser instrument model LA950 at the Sedimentology Laboratory of the University of Bucharest. The samples were pretreated with H_2O_2 for removal of organic matter, HCl at pH 4 for removal of carbonates, and



dispersed with sodium hexametaphosphate. Samples were additionally dispersed using the ultrasonic device of the Horiba instrument. The grain-size distribution was calculated using the Mie theory, using refractive indices of 1.33 (fluid) and 1.54 (sample) (Mie, 1908).

4. Results and Discussion

4.1. Unmixing Backfield IRM Curves and Nonlinear Preisach Maps

4.1.1. NMF Unmixing Model for Backfield IRM Curves

The loess-paleosol backfield IRM data from Lunca can be explained using a two-EM mixing model, with a R^2 coefficient of determination of 0.99924 (Figure 2). A three-EM mixing model, with a R^2 coefficient of determination of 0.99936, provides only a minor improvement in terms of the model fit. The remanence gradient curves of the two EMs have median coercivities of 21 mT (EM1) and 40 mT (EM2) (Figure 2a). Typical examples of individual representative samples fits are presented in Figures 2c and 2e. For comparison, the NMF unmixing end-members and individual representative samples fits for Costineşti section are displayed in Figures 2d and 2f (Necula et al., 2015). Coefficients of determination between 0.998 and 0.999 for all the Lunca and Costineşti samples prove that the backfield IRM data are well described by the NMF model.

The variation of the relative contributions of the two EMs with depth is plotted in Figure 3a. Similar relative contributions resulted from NMF unmixing of backfield IRM curves from Costineşti section (Figure 3b) (Necula et al., 2015). The absolute contributions of EM1 and EM2, along with the granulometry and magnetic susceptibility, $\chi_{\rm lf}$ (Constantin et al., 2015) for Lunca section are plotted in Figure 3c. EM1 and the pedogenic grain size fraction (<5 µm) correlate positively with $\chi_{\rm lf}$, whereas EM2 and the airborne fraction (>16 µm) correlate negatively with $\chi_{\rm lf}$.

4.1.2. Nonlinear Preisach Maps

The nonlinear Preisach maps of representative loess and paleosol samples from both loess sections are displayed in Figure 4. The amplitude of the maps of all paleosols are dominated by a non-interacting SD ridge along the main diagonal (upper left to bottom right) of the diagram (Church et al., 2016). The SD ridge extends to 1 T, with the signal intensity attaining maximum between 20 and 30 mT. In addition to the SD ridge, there is a weaker amplitude contribution to the background of the Preisach map, showing spreading about the diagonal, with an asymmetrical, downward shift indicating the presence of PSD/MD particles (Church et al., 2016).

The nonlinear Preisach maps of the loess samples contain the same features, but showing a higher intensity of the background signature, relative to the SD ridge. The weak signal extension along the diagonal, with coercivities up to 1T in both loess and paleosol samples, suggests an antiferromagnetic contribution.

4.1.3. Unmixing of Nonlinear Preisach Maps

The variability in the nonlinear Preisach maps for the Lunca and Costineşti data sets is accounted for by two principal components (PCs), which explain more than 99.99% of the variance in both sections (Figure S3 in Supporting Information S1). The EMs and their relative contributions are depicted in Figure 5 (Lunca) and Figure 6 (Costineşti). Both sections present virtually identical EMs. EM1 (Figures 5a and 6a) is dominated by the SD diagonal ridge with low coercivities (maximum between 20 and 30 mT) (Church et al., 2016). EM2 (Figures 5b and 6b) is dominated by the PSD/MD signature (Church et al., 2016). The range of coercivities spanned by EM2 is relatively large (from a few mT to 1 T). In addition to the coarse grain signature, there is a strong signal along the diagonal, at coercivities of 50–100 mT, indicating a contribution from SD particles that are different from the ones dominating EM1. There is an additional, weaker, high coercivity signal along the diagonal, at coercivities an antiferromagnetic contribution of non-interacting SD particles.

The effects of measurements noise on the calculated nonlinear Preisach maps are clearly visible in Figure 4. Calculating statistical significance levels would be necessary to distinguish between signal and noise (see e.g., Heslop & Roberts, 2012). However, the use of SVD as a first step in the unmixing procedure presents the advantage of performing denoising on the EMs, compared to individual Preisach maps (Harrison et al., 2018). This smoothing effect can be seen for EM2 (Figures 5b and 6b), which shows a negative region below the main





Figure 2. Negative matrix factorization isothermal remanent magnetization unmixing results for Lunca (a,c, e) and Costineşti (b, d, f): (a) and (b) Coercivity distributions for EM1 (21 mT median coercivity) and EM2 (40 mT median coercivity) for Lunca (this study) and Costineşti (Necula et al., 2015) sections respectively; (c) and (e) Individual representative samples fits of loess and paleosol using the two end members model for Lunca section; (d) and (f) the same as (c) and (e) but for Costineşti section (Necula et al., 2015).

positive signal, between 30 and 100 mT, which is a characteristic feature of PSD/MD magnetite (Church et al., 2016; Egli, 2021; Lascu et al., 2018) that is hard to distinguish otherwise in individual Preisach maps.

The distribution of coercivities from a nonlinear Preisach map can be calculated directly through summation over the magnetization of the remanent Preisach map (Church et al., 2016). The resulting distributions of coercivities for the two EMs show peaks at ~22 mT for EM1 and ~60 mT for EM2 (Figures 5c and 6c) for both sections. The EM1 peak value is similar to that of EM1 from the NMF IRM unmixing, but the distribution is restricted to coercivities <100 mT. The EM2 peak is at a value that is higher than that of EM2 from NMF IRM unmixing. The relative contributions of the two EMs are plotted in Figures 5d and 6d.





Figure 3. Negative matrix factorization isothermal remanent magnetization unmixing abundances of the two EMs: (a) relative abundances for Lunca; (b) relative abundances for Costineşti (Necula et al., 2015); (c) Absolute remanence contributions of the two endmembers, together with magnetic susceptibility, $\chi_{\rm If}$ (Constantin et al., 2015), and the relative abundances of the clay (<5 µm) and airborne (>16 µm) granulometric fractions for Lunca section.

4.2. Origin of the Magnetic Minerals

The NMF unmixing model for the backfield IRM curves for the Lunca section (Figure 2a) shows the presence of two components, with median coercivities of 21 mT (EM1) and 40 mT (EM2). EM1 is interpreted as pedogenic because it covaries with the magnetic susceptibility and has an important contribution in all paleosols, and only a minor contribution in the loess layers (Figures 3a and 3c). Using the same unmixing method, the pedogenic component was also found dominant in the paleosol units from the Dobrogea Plateau at Costineşti and Mircea Vodă (see Figure 2b; Necula et al., 2013, 2015), as well as in the Chaona loess section (Chinese Loess Plateau, Nie et al., 2014). Other IRM deconvolution methods showed similar median coercivities for pedogenic magnetite/



Journal of Geophysical Research: Solid Earth



Figure 4. Normalized nonlinear Preisach maps of representative paleosol and loess samples from Lunca (a) S5 and (b) L1 and Costineşti (c) S3 and (d) L2.

maghemite worldwide (Egli, 2004a, 2004b; Geiss et al., 2008; Hu et al., 2013; Jordanova & Jordanova, 2020; Jordanova et al., 2011; Spassov et al., 2003). The currently accepted interpretation of this component is that it is pedogenic magnetite, with a grain size spanning the SP/SD boundary (20–25 nm) and a wide enough distribution to carry a remanent magnetization. Our IRM unmixing results corroborate previous findings that coercivity analysis of modern soils and paleosols highlights a pedogenic magnetite component which is relatively consistent across different continents and environmental conditions (e.g., Maxbauer et al., 2016a and references therein). EM2 is dominant in loess units, with only minor contributions in palaeosols, and is interpreted as a detrital component. This component has also been found in the Dobrogea Plateau, but with a higher median coercivity: ~60 mT at Mircea Vodă (Necula et al., 2013) and ~50 mT at Costineşti (Figure 2b; Necula et al., 2015). A comparable detrital component has also been reported in Lower Danube Basin loess, as well as in the Chinese Loess Plateau (e.g., Hu et al., 2013; Jordanova et al., 2011; Nie et al., 2014; Spassov et al., 2003). All these studies have interpreted this component as being a detrital magnetite or partially oxidized detrital magnetite (magnetite core with maghemite rim resulting from weathering).

As Church et al. (2016) pointed out, nonlinear Preisach maps are more robust than IRM curve unmixing because they allow both the separation of components with distinct coercivity distributions, and the identification of magnetization processes associated with different magnetic phases found in a bulk sample. The Preisach EM1 coercivity distribution is identical both at Lunca and Costineşti, with a median coercivity of ~22 mT (Figures 5c and 6c). However the EM1 component from backfield IRM NMF unmixing shows a high-coercivity tail which is not present in the corresponding Preisach EM1. The Preisach EM2 distribution has a median coercivity of ~60 mT (Figures 5c and 6c), a higher value than that of EM2 from NMF IRM unmixing, especially for Lunca. These discrepancies may come from using different experimental protocols, number of data points (30 for IRM acquisition vs. 2,500 for Preisach), conditioning field steps, measurement noise, unmixing algorithms, etc (e.g., Heslop, 2015; Heslop & Dillon, 2007). To resolve these discrepancies, we applied the SVD + ICA unmixing method to the IRM backfield curve data sets from both Lunca and Costineşti. The analysis yielded two EMs in both sections, which explain about 99.96% of the variance at Costineşti and 99.98% of the variance at Lunca



Journal of Geophysical Research: Solid Earth





(Figure S4 in Supporting Information S1). The two EMs and their relative contributions are illustrated in Figure 7. At both locations the median coercivity for EM1 is 21 mT and for EM2 is 60 mT, matching closely the median coercivities of the Preisach EMs (Figures 5c and 6c). This analysis provides strong evidence that both sets of EMs describe the same magnetic components. The backfield IRM EM1 is controlled by the same population of grains that controls the Preisach EM1 component. Based on the Preisach maps of EM1 (Figures 5a and 6a), we can determine unambiguously that this population of grains is dominated by low-coercivity non-interacting SD magnetite. Thus, using Preisach unmixing, we can clearly state that EM1 contains a single magnetic population, exclusively of pedogenic origin. On the other hand, Preisach unmixing demonstrates that EM2 does not represent a single magnetic phase, but is comprised of a mixture of PSD/MD and SD magnetite, as well as non-interacting SD hematite, all with the same origin, likely eolian. These multiple components cannot be further extracted because their coercivity spectra overlap substantially (e.g., Egli, 2021; He et al., 2020).

The relative abundances of EM1 and EM2 differ according to the unmixing method used. Thus, NMF IRM unmixing provides higher EM1 abundances (Figures 3a and 3b) than SVD + ICA IRM unmixing (Figures 7b and 7d), and Preisach maps unmixing (Figures 5d and 6d). In case of nonparametric unmixing methods the shape of EMs and the resulting relative abundance are closely related (He et al., 2020; Heslop, 2015). Thus, the main cause of discrepancy between the results of NMF and SVD + ICA backfield IRM unmixing is related to the specific protocols each method uses: different shapes of the EM coercivity distributions provided by each unmixing protocol will give rise simultaneously to different abundances. For example, NMF considers only non-negative, sum-to-1 solutions for the mixing matrix, while SVD + ICA does not have such constraints. In case of a two EM model, considering only sum-to-1 solutions for the mixing matrix forces the relative contributions of the EMs to take 0 and 1 values (e.g., Figure 3a). This implies that the EMs themselves are contained in the data set, assumption that is not always realistic. More commonly, all samples are mixtures between the two EMs in



Journal of Geophysical Research: Solid Earth



Figure 6. Nonlinear Preisach maps unmixing results for Costineşti: (a) EM1 nonlinear Preisach map; (b) EM2 nonlinear Preisach map; (c) Coercivity distributions for EM1 (22 mT median coercivity) and EM2 (65 mT median coercivity); (d) Relative abundances of the two endmembers in loess (L) and paleosol (S) units.

different proportions, with none of the samples representing a pure EM. On the other hand, SVD + ICA takes as EMs the most non-Gaussian solutions without imposing sum to 1 condition. Therefore, it is worth remind readers to carefully take into consideration the sensitivity of the unmixing methods in determining the EMs reflecting some inherent uncertainty associated to each specific analysis.

On the other hand the coercivity distributions from Preisach maps unmixing (Figures 5c and 6c) and SVD + ICA backfield IRM unmixing match closely (Figures 7a and 7c). Furthermore, the EM proportions from both methods show a strong linear dependence ($R^2 = 0.99$ for Costineşti and $R^2 = 0.96$ for Lunca) and slopes close to 1 for both sections (Figure 8). This reinforces the fact that the backfield IRM coercivity distributions and Preisach maps capture the same information. Small differences between the results from IRM curves and Preisach maps could be due to the fact that the measurements were done on sister samples, that the backfield IRMs were measured for fields from 1 mT to 1 T, while Preisach maps contain a wider range of fields (from 0.1 mT to 1 T), and that there was a significant difference between the number of backfield IRM samples and Preisach maps samples (168 and 131 vs. 13 and 11 respectively).

However, results from both unmixing methods provide the same message: loess layers are dominated by detrital magnetic particles of eolian origin, while paleosols are dominated by the pedogenic fraction consisting mainly in non-interacting SD magnetite/maghemite. Some exceptions are paleosol S1 from Lunca and loess units L4 and L5 from Costineşti, which exhibit a more even proportion of EMs. The Lunca S1 paleosol has probably not been developed enough to lead to a predominance of the pedogenic fraction over the detrital one. This is supported by the bulk magnetic susceptibility and pedogenic fraction (<5 μ m) variations (Figure 3c) which contain the lowest values in this unit at Lunca. The Costineşti L4 and L5 loess layers, on the other hand, have a significant EM1





Figure 7. SVD + ICA isothermal remanent magnetization unmixing results for Lunca (a–b) and Costineşti (c–d): Coercivity distributions for EM1 (21 mT median coercivity) and EM2 (60 mT median coercivity) (a and c) and relative abundances of the two endmembers in loess (L) and paleosol (S) units (b and d).

fraction suggesting that these units are affected by pedogenesis. This is also supported by all the concentrationdependent magnetic parameters from Costineşti, which, in these units, show values as high as in the S0, S1 and S2 paleosols (Necula et al., 2015). Therefore, Preisach maps can be used as a straightforward tool for explaining the IRM coercivity distributions in terms of domain state for the main ferromagnetic minerals.



Figure 8. Comparison of relative abundances resulting from SVD + ICA isothermal remanent magnetization and nonlinear Preisach maps unmixing for Lunca (a) and Costineşti (b). Shaded regions represent 95% confidence levels.



4.3. Comparison to Standard Unmixing Techniques

4.3.1. Single Sample Parametric Backfield IRM Modeling

The unmixing results for parametric remanence decomposition using MaxUnmix (Maxbauer et al., 2016b) are depicted in Figure 9. For S0, the remanence gradient has a single peak and it is almost symmetric and very narrow suggesting overlap between coercivity components (Figure 9). We tested the parametric unmixing in two steps. Firstly, we checked if decomposition depends on the type of model distributions used to fit the data (Egli, 2003; Zhao et al., 2018). MaxUnmix application includes two of these namely SN and simple Gaussian (G) distributions. At least three components are needed to fit the coercivity distribution for both types of curves (Figures 9a and 9b), keeping the same initial parameters for both. For the first two components the median coercivity remains the same (4 and 22 mT using SN, and 5 and 21 mT using G respectively). The third component, however, has different coercivities in the two cases (88 mT for SN and 70 mT for G). Moreover, the amplitude of the components varies depending on the type of curve used, the most noticeable being the second component (Figures 8a and 8b). Secondly, we tested the influence of initial conditions on the unmixing results. Therefore, we changed the coercivity of the third component to ~ 100 mT and only slightly for the first two components. The new fitting solution seems to be as good as the previous one showing that a third component with a median coercivity of 102 mT is also valid (Figures 9c and 9d). However, the first component's median coercivity is modified substantially when using G, showing the dependence of curve type. Overall, only the second component (21-24 mT median coercivity) is more or less stable, regardless of initial conditions and types of functions used for fitting the data.

The interpretation of the three components can be ambiguous. The first component may be interpreted as having several origins: it could represent MD or SP particles signatures (Egli, 2004a, 2004b), could be the result of magnetostatic interactions between magnetic particles (Egli, 2004a, 2004b; Heslop et al., 2002, 2004), could be simply a filler because the model distributions cannot mimic the real data (Egli, 2003; Zhao et al., 2018), or any combination of the above. The third component can be interpreted as partially oxidized magnetite if coercivity is in the 70–88 mT range, but can also be interpreted as maghemite and/or hematite if its coercivity is ~100 mT (e.g., Spassov et al., 2003). For the second component (21–24 mT) the interpretation is more or less straightforward: low coercivity pedogenic magnetite most probably in SD state (e.g., Maxbauer et al., 2016a).

For L1 we also obtained three components without constraining the coercivity values. For the SN model the median coercivities of the components are 7, 24, and 78 mT (Figure 9e), while for the G model they are 8, 25, and 78 mT (Figure 9f), matching the components obtained for the unconstrained S0 models. However, the third component is consistent regardless of the type of theoretical curves used to fit the data, and can be interpreted as partially oxidized magnetite. The amplitude of this component is larger compared to the paleosol sample, where the second component has the largest amplitude, indicating a dominance of particles of eolian origin.

Given the ambiguity of the first component, there are probably only two major components that contribute to the loess and paleosol representative samples from Lunca: a pedogenic component and one of eolian origin. The pedogenic component is similar to that provided by the NMF and SVD + ICA methods, whereas the eolian one has notably higher coercivity. If additional information about the sample (e.g., domain state, magnetic mineralogy, magnetostatic interactions) is lacking it is challenging to interpret these solutions. Therefore, using Pre-isach maps we can straightforwardly interpret the coercivity components given by any of these remanence curves unmixing methods (parametric and/or non parametric).

4.3.2. FORC-PCA Unmixing

FORC-PCA yielded two EMs along the first PC (eigenvector 0), which explains nearly 98% of the variability in the data set (Figures 10a and 10b). EM1 (Figure 10c) has a signature dominated by a low coercivity central ridge (maximum amplitude at about 10 mT), suggesting the presence of non-interacting SD particles. The background EM1 signature could indicate strong magnetostatic interactions or the presence of a PSD/MD component. EM2 is dominated by a PSD/MD signal (Figure 10d), with a weak central ridge signature. The FORC diagrams of the EMs suggest they are likely mixtures of pedogenic and eolian components, indicating that the FORC-PCA separation is not optimal. Moreover, there is no high coercivity component captured. For this type of high-resolution measurement the results are limited to a narrow range of coercivities and interaction fields due to





Figure 9. Parametric unmixing of backfield isothermal remanent magnetization curves for representative samples of paleosol (S0) and loess (L1) from Lunca section: (a) S0 modeling using skewed normal curves; (b) S0 modeling using Gaussian curves; (c) and (d) the same as (a) and (b) with modified initial parameters; (e) and (f) the same as (a) and (b) for L1. Gray dots represent the data and yellow line represents the modeled coercivity distribution. Shaded areas represent 95% confidence intervals associated with each component.

10.1029/2024JB029004



Journal of Geophysical Research: Solid Earth



Figure 10. FORC-PCA for Lunca section. (a) Scree plot; (b) Score plot, including endmember (EM) location and contours for "feasibility metrics"; (c) first-order reversal curve (FORC) EM1; (d) FORC EM2; (e) Relative contributions of the two EMs.

the linear nature of data collection. In our case, the FORC distribution is limited to a few tens of mT along both the coercivity and interaction field axes.

Preisach maps on the other hand, due to the logarithmic field steps, allow the inspection of large ranges of coercivities and interaction fields (on the order of 1-2 T). This, coupled with the fact that measurements are done in zero field, allow the much better expression of the high-coercivity antiferromagnetic signal. Preisach maps are also not smoothed, thus avoiding artifacts associated to this post processing operation. FORC unmixing results cannot be directly compared to the remanence-based unmixing results because, FORCs capture a combination of induced, transient, remanent, and thermal relaxation magnetizations. An effect of this is that FORC EM1 maximum coercivity has a lower value (10 mT) than EM1 maximum coercivity from Preisach maps unmixing (~21 mT). Moreover the Preisach unmixing provides better separation of the true pedogenic and eolian components. Preisach EM1 is easily identifiable as noninteracting SD particles (dominance of ridge signal) of pedogenic origin, while Preisach EM2 contains a mix of particles of eolian origin (PSD/MD + higher coercivity SD ferrimagnetic, as well as antiferromagnetic contributions). Finally the entire FORC-PCA procedure (collecting data, processing FORCs, and setting up the FORC-PCA) is at least as time consuming as the Preisach unmixing. All these indicate Preisach maps as the most suitable to help the interpretation of IRM unmixing in term of domain states and interactions, at least for loess-paleosol deposits.

4.4. Pedogenesis and Provenance in Lower Danube Loess-Paleosol Deposits

The two loess sections were developed under different climatic conditions. At Lunca, pedogenesis intensity decreases gradually from S5 to S1, suggesting progressively drier interglacial conditions with time (Figure 3c). Obreht et al. (2016) found a similar aridification pattern in the Stalač loess section in central Serbia, documenting a transition from Mediterranean climate to more continental conditions in the Balkans from MIS 9 to present. The same aridification trend was described in the Zeemun loess sequence in the Middle Danube Basin starting with MIS 11 (Laag et al., 2021). A similar trend of increased continentality over time is also observed around the Azov Sea, suggesting a supra-regional climatic trend (Liang et al., 2016; Obreht et al., 2016). The Lunca section documents the highest intensity of pedogenesis in the S5 paleosol suggesting that this aridification trend could have started in the area as early as MIS13-15, 500–600 ka ago (Figure 3c).

At Costineşti, a significant shift from wetter to drier conditions occurred only ~300 ka ago (for Costineşti age model please refer to Constantin et al. (2013). Before that, Mediterranean climate was dominant (600–300 ka), with pedogenesis occurring in loess layers L5 and L4 in addition to the paleosols (S5 to S3). From 300 ka ago to present, continental climate conditions prevailed during interglacials (S2 to S0 paleosols) (Buggle et al., 2009, 2013, 2014; Necula et al., 2015). Moreover, bulk magnetic parameters recorded at Costineşti show systematically higher values in palaeosols than at Lunca, suggesting a more intense pedogenesis at Costineşti since 300 ka (Necula et al., 2015).

Preisach maps unmixing revealed identical EM1 in both sections irrespective of the degree of pedogenesis and the evolution of climate during the last 700 ka. This implies that the pedogenesis produce the same type of magnetic mineral, namely non interacting SD magnetite/maghemite, with the same coercivity distributions. The variations in EM1 abundances at the two sites must then reflect the intensity of pedogenesis, which is in turn controlled by climate, with more magnetite being produced during wetter conditions.

The source areas for the loess deposits in the Middle and Lower Danube Basin, as well as for the Black Sea shore, are predominantly local (Lehmkuhl et al., 2021). In the Lower Danube Basin, the main source of dust is thought to be the very alluvium of the Danube River (Fenn et al., 2022; Lehmkuhl et al., 2021). In the Dobrogea Plateau and Black Sea shore, the main source of dust is the continental shelf of the Black Sea, which is exposed during glacials as a result of sea level fall (Hoyle et al., 2021; Jipa, 2014). Loess granulometry shows a systematic finer grain size at Costineşti compared to Lunca (e.g., more than 4 times finer in unit L5), supporting the model of different source areas for the two locations (Figure 11).

Finer dust could also imply transport from greater distances. Indeed, it has been documented that additional sources of dust for this area are Ukrainian glaciofluvial sediments (Buggle et al., 2008), and the Aral and Caspian arid lands (Stephens et al., 2003). Preisach EM2, which is of eolian origin, can be used to characterize the magnetic population of the source area, as well as transport processes. EM2 shows identical magnetic mineral content both at Costineşti and Lunca, suggesting that the same magnetic components are transported and deposited in the region irrespective of the dust provenance and transport processes. Moreover SD and/or PSD magnetite is probably oxidized, which gives its higher coercivity compared to paleosols (e.g., Ge et al., 2014; Özdemir & Dunlop, 2010; van Velzen & Dekkers, 1999).





Figure 11. Median grain size for Lunca (a) and Costineşti (b) loess (L) and paleosol (S) units.

5. Conclusions

Our rock magnetic investigation of two loess-paleosol sections from Southeastern Europe showed that the combination of non-parametric backfield IRM unmixing and Preisach maps unmixing can be successfully used for identifying magnetic components and processes in loess-paleosol deposits, providing a considerable advantage in their interpretation. Our findings can be summarized as follows.

- The nonparametric unmixing method can provide more robust insight into the difference in magnetic properties of constituent magnetic components than parametric unmixing method. However, uncertainty still remains as the number of EMs, their shapes and thus abundances may still depend on the unmixing method used. Preisach maps provide useful information on the internal structure of the EMs that is critical for interpreting the mineral phases and domain states of the components.
- Our results demonstrate that EM1 is dominated by pedogenic, non-interacting SD magnetite/maghemite with low coercivity. EM2 is a mixture of magnetic phases comprising PSD/MD and SD magnetite/maghemite with higher coercivity than EM1, as well as non-interacting SD hematite, all of eolian origin.
- Pedogenesis produces the same type of magnetic mineral, namely non-interacting SD magnetite/maghemite, with the same coercivity distribution regardless of location. Since this coercivity component has been consistently found worldwide in loessic soils, nonlinear Preisach measurements can provide a way to unambiguously diagnose its presence.
- The eolian EM contains both coarse (PSD/MD) and fine (SD of higher coercivity than EM1 most probably due to low temperature oxidation) ferrimagnetic particles, as well as an antiferromagnetic component. Thus, nonlinear Preisach maps allow the identification of the domain states of the mineral phases making up this complex composition. The same EM2 in the two loess sequences indicates that the same magnetic dust fraction is transported and deposited irrespective of provenance and transport processes.



21699356, 2024, 6, Downloaded from I

ary.wiley

.com/doi/10.1029/2024JB029004 by Cochran

Federation, Wiley Online Library on [11/12/2024]. See

the

Term

Online

use; O/

rned by the applicable Crea

• There are drawbacks to using non-linear Preisach maps. Because they are based on remanence curves, Preisach maps will not capture the superparamagnetic component of pedogenic enhancement, which may be significant. The measurements are much more time consuming than IRM curves but not than FORCs and, like FORCs, they are generally impractical for a large number of samples. Non-linear Preisach measurements are thus ideal for characterization of representative samples, at a resolution that captures all the magnetic features of interest.

Data Availability Statement

All figures were made with Matplotlib version 3.6.1 (Caswell et al., 2022; Hunter, 2007), available under the Matplotlib license and Inkscape version 1.2 (Knoth et al., 2022) under GNU General Public License, version 3. Map from Figure 1 was created through Cartopy (Elson et al., 2022) licensed under LGPL version 3 or later and Inkscape v1.2 (Knoth et al., 2022). All unmixing were performed using Hyperspy module version 1.6.5 (de la Peña et al., 2021) licensed under GNU General Public License v3.0. The magnetic and granulometry data used in this study are available at Zenodo (Necula et al., 2024), with open access, licensed under Creative Commons Attribution 4.0 International. The Jupyter Notebooks to execute the analysis presented in the paper are hosted at GitHub and preserved at Zenodo (Necula et al., 2024), licensed under Creative Commons Attribution 4.0 International.

References

- Abrajevitch, A., Font, E., Florindo, F., & Roberts, A. P. (2015). Asteroid impact vs. Deccan eruptions: The origin of low magnetic susceptibility beds below the Cretaceous–Paleogene boundary revisited. *Earth and Planetary Science Letters*, 430, 209–223. https://doi.org/10.1016/j.epsl. 2015.08.022
- Abrajevitch, A., Hori, R. S., & Kodama, K. (2013). Rock magnetic record of the Triassic-Jurassic transition in pelagic bedded chert of the Inuyama section, Japan. *Geology*, 41(7), 803–806. https://doi.org/10.1130/g34343.1
- Abrajevitch, A., Van der Voo, R., & Rea, D. K. (2009). Variations in relative abundances of goethite and hematite in Bengal Fan sediments: Climatic vs. diagenetic signals. *Marine Geology*, 267(3–4), 191–206. https://doi.org/10.1016/j.margeo.2009.10.010
- Bai, F., Chang, L., Berndt, T. A., & Pei, Z. (2021). Micromagnetic calculations of the effect of magnetostatic interactions on isothermal remanent magnetization curves: Implications for magnetic mineral identification. Journal of Geophysical Research: Solid Earth, 126(7), e2021JB022335. https://doi.org/10.1029/2021jb022335
- Balescu, S., Jordanova, D., Forget Brisson, L., Hardy, F., Huot, S., & Lamothe, M. (2019). Luminescence chronology of the northeastern Bulgarian loess-paleosol sequences (Viatovo and Kaolinovo). *Quaternary International*, 552, 15–24. https://doi.org/10.1016/j.quaint.2019. 04.020
- Buggle, B., Glaser, B., Z"oller, L., Hambach, U., Markovi'c, S. B., Glaser, I., & Gerasimenko, N. P. (2008). Geochemical characterization and origin of Southeastern and Eastern European loesses (Serbia, Romania, Ukraine). *Quaternary Science Reviews*, 27(9–10), 1058–1075. https:// doi.org/10.1016/j.quascirev.2008.01.018
- Buggle, B., Hambach, U., Glaser, B., Gerasimenko, N., Marković, S. B., Glaser, I., & Zöller, L. (2009). Stratigraphy and spatial and temporal paleoclimatic trends in East European loess paleosol sequences. *Quaternary International*, 196(1–2), 86–106. https://doi.org/10.1016/j.quaint. 2008.07.013
- Buggle, B., Hambach, U., Kehl, M., Marković, S. B., Zöller, L., & Glaser, B. (2013). The progressive evolution of a continental climate in SE-Central European lowlands during the Middle Pleistocene recorded in loess paleosol sequences. *Geology*, 41(7), 771–774. https://doi.org/10. 1130/g34198.1
- Buggle, B., Hambach, U., Müller, K., Zöller, L., Marković, S. B., & Glaser, B. (2014). Iron mineralogical proxies and Quaternary climate change in SE-European loess–paleosol sequences. *Catena*, 117, 4–22. https://doi.org/10.1016/j.catena.2013.06.012
- Caswell, T. A., Lee, A., Droettboom, M., de Andrade, E. S., Hoffmann, T., Klymak, J., et al. (2022). matplotlib/matplotlib: Rel: V3.6.1 (v3.6.1) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.7162185
- Chen, L., Heslop, D., Roberts, A. P., Chang, L., Zhao, X., McGregor, H. V., et al. (2017). Remanence acquisition efficiency in biogenic and detrital magnetite and recording of geomagnetic paleointensity. *Geochemistry, Geophysics, Geosystems*, 18(4), 1435–1450. https://doi.org/10. 1002/2016gc006753
- Church, N. S., Fabian, K., & McEnroe, S. A. (2016). Nonlinear Preisach maps: Detecting and characterizing separate remanent magnetic fractions in complex natural samples. Journal of Geophysical Research: Solid Earth, 121(2), 8373–8395. https://doi.org/10.1002/2016jb013465
- Constantin, D., Begy, R., Vasiliniuc, S., Panaiotu, C., Necula, C., Codrea, V., & Timar-Gabor, A. (2013). High-resolution OSL dating of the Costineşti section (Dobrogea, SE Romania) using fine and coarse quartz. *Quaternary International*, 334, 20–29. https://doi.org/10.1016/j. quaint.2013.06.016
- Constantin, D., Cameniță, A., Panaiotu, C., Necula, C., Codrea, V., & Timar-Gabor, A. (2015). Fine and coarse-quartz SAR-OSL dating of last glacial loess in southern Romania. *Quaternary International*, 357, 33–43. https://doi.org/10.1016/j.quaint.2014.07.052
- Dekkers, M. J. (2012). End-member modelling as an aid to diagnose remagnetization: A brief review. *Geological Society, London, Special Publications*, 371(1), 253–269. https://doi.org/10.1144/SP371.12
- de la Peña, F., Berger, M.-H., Hochepied, J.-F., Dynys, F., Stephan, O., & Walls, M. (2011). Mapping titanium and tin oxide phases using EELS: An application of independent component analysis. *Ultramicroscopy*, 111(2), 169–176. https://doi.org/10.1016/j.ultramic.2010.10.001
- de la Peña, F., Prestat, E., Fauske, V. D., Burdet, P., Lähnemann, J., Furnival, T., et al. (2021). hyperspy/hyperspy: Release v1.6.5 (v1.6.5) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.5608741
- Egli, R. (2003). Analysis of the field dependence of remanent magnetization curves. *Journal of Geophysical Research*, 108(B2), 2081. https://doi.org/10.1029/2002JB002023

Acknowledgments

CN, CP and DG were supported by grant EEA-RO-NO2018-0126, 2019–2023. IL was supported by National Science Foundation Grant EAR-2044506. We thank the editor, the expert associate editor and the reviewers for their valuable comments and suggestions which significantly improved the manuscript. Open access publishing facilitated by Anelis Plus (the official name of "Asociatia Universitatilor, a Institutelor de Cercetare – Dezvoltare si a Bibliotecilor Centrale Universitare din Romania"), as part of the Wiley - Anelis Plus agreement.



- Egli, R. (2004a). Characterization of individual rock magnetic components by analysis of remanence curves, 1. Unmixing natural sediments. *Studia Geophysica et Geodaetica*, 48(2), 391–446. https://doi.org/10.1023/B:SGEG.0000020839.45304.6d
- Egli, R. (2004b). Characterization of individual rock magnetic components by analysis of remanence curves. 2. Fundamental properties of coercivity distributions. *Physics and Chemistry of the Earth*, 9(13–14), 851–867. https://doi.org/10.1016/j.pce.2004.04.001
- Egli, R. (2013). VARIFORC: An optimized protocol for calculating non-regular first-order reversal curve (FORC) diagrams. *Global and Planetary Change*, 110, 302–320. https://doi.org/10.1016/j.gloplacha.2013.08.003
- Egli, R. (2021). Magnetic characterization of geologic materials with first-order reversal curves. In V. Franco & B. Dodrill (Eds.), Magnetic measurement techniques for materials characterization. Springer. https://doi.org/10.1007/978-3-030-70443-8_17
- Elson, P., de Andrade, E. S., Lucas, G., May, R., Hattersley, R., Campbell, E., et al. (2022). SciTools/cartopy: V0.21.1 (v0.21.1) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.7430317
- Fabian, K., Shcherbakov, V. P., Kosareva, L., & Nourgaliev, D. (2016). Physical interpretation of isothermal remanent magnetization endmembers: New insights into the environmental history of Lake Hovsgul, Mongolia. *Geochemistry, Geophysics, Geosystems*, 17(11), 4669–4683. https://doi.org/10.1002/2016GC006506
- Fenn, K., Millar, I. L., Durcan, J. A., Thomas, D. S. G., Banak, A., Marković, S. B., et al. (2022). The provenance of Danubian loess. *Earth-Science Reviews*, 226, 103920. https://doi.org/10.1016/j.earscirev.2022.103920
- Fitzsimmons, K. E., Marković, S. B., & Hambach, U. (2012). Pleistocene environmental dynamics recorded in the loess of the middle and lower Danube basin. *Quaternary Science Reviews*, 41, 104–118. https://doi.org/10.1016/j.quascirev.2012.03.002
- Ge, K., Williams, W., Liu, Q., & Yu, Y. (2014). Effects of the core-shell structure on the magnetic properties of partially oxidized magnetite grains: Experimental and micromagnetic investigations. *Geochemistry, Geophysics, Geosystems*, 15(5), 2021–2038. https://doi.org/10.1002/ 2014gc005265
- Geiss, C. E., Egli, R., & Zanner, C. W. (2008). Direct estimates of pedogenic magnetite as a tool to reconstruct past climates from buried soils. Journal of Geophysical Research, 113(B11), B11102. https://doi.org/10.1029/2008JB005669
- Harrison, R., Zhao, X., Hu, P., Sato, T., Heslop, D., Muxworthy, A., et al. (2019). Simulation of remanent, transient, and induced first-order reversal curve (FORC) diagrams for interacting particles with uniaxial, cubic, and hexagonal anisotropy. *Journal of Geophysical Research: Solid Earth*, 124(12), 12404–12429. https://doi.org/10.1029/2019JB018050
- Harrison, R. J., & Feinberg, J. M. (2008). FORCinel: An improved algorithm for calculating first-order reversal curve distributions using locally weighted regression smoothing. *Geochemistry, Geophysics, Geosystems*, 9(5), Q05016. https://doi.org/10.1029/2008GC001987
- Harrison, R. J., Muraszko, J., Heslop, D., Lascu, I., Muxworthy, A. R., & Roberts, A. P. (2018). An improved algorithm for unmixing first-order reversal curve diagrams using principal component analysis. *Geochemistry, Geophysics, Geosystems*, 9(5), 1595–1610. https://doi.org/10. 1029/2018gc007511
- He, K., Zhao, X., Pan, Y., Zhao, X., Qin, H., & Zhang, T. (2020). Benchmarking component analysis of remanent magnetization curves with a synthetic mixture series: Insight into the reliability of unmixing natural samples. https://doi.org/10.1029/2020JB020105
- Heslop, D. (2015). Numerical strategies for magnetic mineral unmixing. *Earth-Science Reviews*, 150, 256–284. https://doi.org/10.1016/j.earscirev.2015.07.007
- Heslop, D., Dekkers, M. J., Kruiver, P. P., & Van Oorschot, I. H. M. (2002). Analysis of isothermal remanent magnetization acquisition curves using the expectation-maximization algorithm. *Geophysical Journal International*, 148(1), 58–64. https://doi.org/10.1046/j.0956-540x.2001. 01558.x
- Heslop, D., & Dillon, M. (2007). Unmixing magnetic remanence curves without a priori knowledge. *Geophysical Journal International*, 170(2), 556–566. https://doi.org/10.1111/j.1365-246x.2007.03432.x
- Heslop, D., McIntosh, G., & Dekkers, M. J. (2004). Using time- and temperature-dependent Preisach models to investigate the limitations of modelling isothermal remanent magnetization acquisition curves with cumulative log Gaussian functions. *Geophysical Journal International*, 157(1), 55–63. https://doi.org/10.1111/j.1365-246X.2004.02155.x
- Heslop, D., & Roberts, A. P. (2012). Estimation of significance levels and confidence intervals for first-order reversal curve distributions. Geochemistry, Geophysics, Geosystems, 13(5), Q12Z40. https://doi.org/10.1029/2012GC004115
- Hoyle, T. M., Bista, D., Flecker, R., Krijgsman, W., & Sangiorgi, F. (2021). Climate-driven connectivity changes of the Black Sea since 430 ka: Testing a dual palynological and geochemical approach. *Palaeogeography, Palaeoclimatology, Palaeoecology, 561*, 110069. https://doi.org/ 10.1016/j.palaeo.2020.110069
- Hu, P., Heslop, D., Viscarra Rossel, R. A., Roberts, A. P., & Zhao, X. (2020). Continental scale magnetic properties of surficial Australian soils. *Earth-Science Reviews*, 203, 103028. https://doi.org/10.1016/j.earscirev.2019.103028
- Hu, P., Oda, H., Zhao, X., Harrison, R. J., Heslop, D., Sato, T., et al. (2021). Assessment of magnetic techniques for understanding complex mixtures of magnetite and hematite: The Inuyama red chert. *Journal of Geophysical Research: Solid Earth*, 126(1), e2020JB019518. https:// doi.org/10.1029/2020JB019518
- Hu, P. X., Liu, Q. S., Heslop, D., Roberts, A. P., & Jin, C. S. (2015). Soil moisture balance and magnetic enhancement in loess–paleosol sequences from the Tibetan Plateau and Chinese Loess Plateau. *Earth and Planetary Science Letters*, 409, 120–132. https://doi.org/10.1016/j.epsl.2014. 10.035
- Hu, P. X., Liu, Q. S., Torrent, J., Barron, V., & Jin, C. S. (2013). Characterizing and quantifying iron oxides in Chinese loess/paleosols: Implications for pedogenesis. *Earth and Planetary Science Letters*, 369, 271–283. https://doi.org/10.1016/j.epsl.2013.03.033
- Hunter, J. D. (2007). Matplotlib: A 2D graphics environment. Computing in Science and Engineering, 9(3), 90–95. https://doi.org/10.1109/ MCSE.2007.55
- Hyland, E., Sheldon, N. D., Van der Voo, R., Badgley, C., & Abrajevitch, A. (2015). A new paleoprecipitation proxy based on soil magnetic properties: Implications for expanding paleoclimate reconstructions. *Geological Society of America Bulletin*, 127(7–8), 975–981. https://doi. org/10.1130/B31207.1
- Hyodo, M., Sano, T., Matsumoto, M., Seto, Y., Bradák, B., Suzuki, K., et al. (2020). Nanosized authigenic magnetite and hematite particles in mature-paleosol phyllosilicates: New evidence for a magnetic enhancement mechanism in loess sequences of China. *Journal of Geophysical Research: Solid Earth*, 125(3), e2019JB018705. https://doi.org/10.1029/2019JB018705
- Hyvärinen, A., Karhunen, J., & Oja, E. (2001). Independent component analysis. Wiley Interscience.
- Hyvärinen, A., & Oja, E. (2000). Independent component analysis: Algorithms and applications. *Neural Networks*, 13(4–5), 411–430. https://doi.org/10.1016/s0893-6080(00)00026-5
- Jipa, D. C. (2014). The conceptual sedimentary model of the lower Danube loess basin: Sedimentogenetic implications. *Quaternary International*, 351, 14–24. https://doi.org/10.1016/j.quaint.2013.06.008



- Jordanova, D., Grygar, T., Jordanova, N., & Petrov, P. (2011). Palaeoclimatic significance of Hematite/Goethite ratio in Bulgarian loess-palaeosol sediments deduced by DRS and rock magnetic measurements. In E. Petrovský, E. Herrero-Bervera, T. Harinarayana, & D. Ivers (Eds.), *The Earth's magnetic interior* (pp. 399–412). Springer.
- Jordanova, D., & Jordanova, N. (2020). Diversity and peculiarities of soil formation in eolian landscapes Insights from the mineral magnetic records. Earth and Planetary Science Letters, 531, 115956. https://doi.org/10.1016/j.epsl.2019.115956
- Jordanova, D., & Petersen, N. (1999). Palaeoclimatic record from a loess–soil profile in northeastern Bulgaria—II. Correlation with global climatic events during the pleistocene. *Geophysical Journal International*, 138(2), 533–540. https://doi.org/10.1046/j.1365-246x.1999.00873.x
- Jordanova, N., Jordanova, D., Tcherkezova, E., Popov, H., Mokreva, A., Georgiev, P., & Stoychev, R. (2020). Identification and classification of archaeological materials from Bronze age gold mining site Ada Tepe (Bulgaria) using rock magnetism. *Earth and Space Science Open Archive*. https://doi.org/10.1002/essoar.10503935.1
- Just, J., Dekkers, M. J., von Dobeneck, T., van Hoesel, A., & Bickert, T. (2012). Signatures and significance of aeolian, fluvial, bacterial and diagenetic magnetic mineral fractions in Late Quaternary marine sediments off Gambia, NW Africa. *Geochemistry, Geophysics, Geosystems*, 13(9), Q0AO02. https://doi.org/10.1029/2012GC004146

Knoth, A., Masár, I., Sharma, A., van Lierop, D., Hirth, J., Petrof, M., et al. (2022). https://inkscape.org/release/inkscape-1.2/

- Kruiver, P. P., Dekkers, M. J., & Heslop, D. (2001). Quantification of magnetic coercivity components by the analysis of acquisition curves of isothermal remanent magnetisation. *Earth and Planetary Science Letters*, 189(3–4), 269–276. https://doi.org/10.1016/S0012-821X(01) 00367-3
- Laag, C., Hambach, U., Zeeden, C., Lagroix, F., Guyodo, Y., Veres, V., et al. (2021). A detailed paleoclimate proxy record for the middle Danube Basin over the last 430 kyr: A rock magnetic and colorimetric study of the zemun loess-paleosol sequence. *Frontiers of Earth Science*, 9, 600086. https://doi.org/10.3389/feart.2021.600086
- Larrasoaña, J. C., Pey, J., Zhao, X., Heslop, D., Mochales, T., Mata, P., et al. (2021). Environmental magnetic fingerprinting of anthropogenic and natural atmospheric deposition over southwestern Europe. Atmospheric Environment, 261, 118568. https://doi.org/10.1016/j.atmosenv.2021. 118568
- Lascu, I., Einsle, J. F., Ball, M. R., & Harrison, R. J. (2018). The vortex state in geologic materials: A micromagnetic perspective. Journal of Geophysical Research: Solid Earth, 123(9), 7285–7304. https://doi.org/10.1029/2018jb015909
- Lascu, I., Harrison, R. J., Li, Y., Muraszko, J., Channell, J. E., Piotrowski, A. M., & Hodell, D. A. (2015). Magnetic unmixing of first-order reversal curve diagrams using principal component analysis. *Geochemistry, Geophysics, Geosystems*, 16(9), 2900–2915. https://doi.org/10. 1002/2015GC005909
- Lehmkuhl, F., Nett, J. J., Pötter, S., Schulte, P., Sprafke, T., Jary, Z., et al. (2021). Loess landscapes of Europe Mapping, geomorphology, and zonal differentiation. *Earth-Science Reviews*, 215, 103496. https://doi.org/10.1016/j.earscirev.2020.103496
- Li, J., Liu, Y., Liu, S., Roberts, A. P., Pan, H., Xiao, T., & Pan, Y. (2020). Classification of a complexly mixed magnetic mineral assemblage in Pacific Ocean surface sediment by electron microscopy and supervised magnetic unmixing. *Frontiers of Earth Science*, 8, 609058. https://doi. org/10.3389/feart.2020.609058
- Liang, Y., Yang, T., Velichko, A. A., Zeng, B., Shi, P., Wang, L., et al. (2016). Paleoclimatic record from Chumbur-Kosa section in Sea of Azov region since marine isotope stage 11. Journal of Mountain Science, 13(6), 985–999. https://doi.org/10.1007/s11629-015-3738-9
- Lisiecki, L. E., & Raymo, M. E. (2005). A Pliocene-Pleistocene stack of 57 globally distributed benthic δ¹⁸O records. *Paleoceanography*, 20(1), PA1003. https://doi.org/10.1029/2004PA001071
- Marković, S. B., Stevens, T., Kukla, G. J., Hambach, U., Fitzsimmons, K. E., Gibbard, P., et al. (2015). Danube loess stratigraphy Towards a pan-European loess stratigraphic model. *Earth-Science Reviews*, 148, 228–258. https://doi.org/10.1016/j.earscirev.2015.06.005
- Maxbauer, D. P., Feinberg, J. M., & Fox, D. L. (2016a). Magnetic mineral assemblages in soils and paleosols as the basis for paleoprecipitation proxies: A review of magnetic methods and challenges. *Earth-Science Reviews*, 155, 28–48. https://doi.org/10.1016/j.earscirev.2016.01.014
- Maxbauer, D. P., Feinberg, J. M., & Fox, D. L. (2016b). MAX UnMix: A web application for unmixing magnetic coercivity distributions. Computers and Geosciences, 95, 140–145, https://doi.org/10.1016/j.cageo.2016.07.009
- Maxbauer, D. P., Feinberg, J. M., Fox, D. L., & Nater, E. A. (2017). Response of pedogenic magnetite to changing vegetation in soils developed under uniform climate, topography, and parent material. *Scientific Reports*, 7(1), 17575. https://doi.org/10.1038/s41598-017-17722-2
- Mie, G. (1908). Beiträge zur Optik trüber Medien, speziell kolloidaler Metallösungen. Annalen der Physik, 330(3), 377–445. https://doi.org/10. 1002/andp.19083300302
- Necula, C., Dimofte, D., & Panaiotu, C. (2015). Rock magnetism of a loess-palaeosol sequence from the western Black Sea shore (Romania). Geophysical Journal International, 202(3), 1733–1748. https://doi.org/10.1093/gjij/ggv250
- Necula, C., Lascu, I., Panaiotu, C., & Gheorghe, D. (2024). Data and software for "Resolving the Interpretation of Magnetic Coercivity Components from Backfield Isothermal Remanence Curves Using Unmixing of Non-linear Preisach Maps: Application to Loess-Paleosol Sequences" by Necula et al [Dataset]. Zenodo Data, V2. https://doi.org/10.5281/zenodo.8385954
- Necula, C., Panaiotu, C., Heslop, D., & Dimofte, D. (2013). Climatic control of magnetic granulometry in the Mircea Vodă loess/paleosol sequence (Dobrogea, Romania). *Quaternary International*, 293, 5–14. https://doi.org/10.1016/j.quaint.2012.03.043
- Nie, J., Garzione, C., Su, Q., Liu, Q., Zhang, R., Heslop, D., et al. (2017). Dominant 100,000 year precipitation cyclicity in a late Miocene lake from northeast Tibet. *Science Advances*, 3(3), e1600762. https://doi.org/10.1126/sciady.1600762
- Nie, J., Zhang, R., Necula, C., Heslop, D., Liu, Q., Gong, L., & Banerjee, S. (2014). Late Miocene–early Pleistocene paleoclimate history of the Chinese Loess Plateau revealed by remanence unmixing. *Geophysical Research Letters*, 41(6), 2163–2168. https://doi.org/10.1002/ 2014gl059379
- Obreht, I., Zeeden, C., Hambach, U., Veres, D., Marković, S. B., Bösken, J., et al. (2016). Tracing the influence of Mediterranean climate on Southeastern Europe during the past 350,000 years. *Scientific Reports*, 6(1), 36334. https://doi.org/10.1038/srep36334org/10.1029/ 2021JB022335
- Obreht, I., Zeeden, C., Hambach, U., Veres, D., Marković, S. B., & Lehmkuhl, F. (2019). A critical reevaluation of palaeoclimate proxy records from loess in the Carpathian Basin. *Earth-Science Reviews*, 190, 498–520. https://doi.org/10.1016/j.earscirev.2019.01.020
- Özdemir, Ö., & Dunlop, D. J. (2010). Hallmarks of maghemitization in low-temperature remanence cycling of partially oxidized magnetite nanoparticles. *Journal of Geophysical Research*, 115, B2. https://doi.org/10.1029/2009jb006756
- Panaiotu, C. G., Panaiotu, E. C., Grama, A., & Necula, C. (2001). Paleoclimatic record from a loess–paleosol profile in Southeastern Romania. *Physics and Chemistry of the Earth*, 26(11–12), 893–898. https://doi.org/10.1016/s1464-1895(01)00138-7
- Roberts, A. P., Chang, L., Heslop, D., Florindo, F., & Larrasoaña, J. C. (2012). Searching for single domain magnetite in the "pseudo-singledomain" sedimentary haystack: Implications of biogenic magnetite preservation for sediment magnetism and relative paleointensity, determinations. Journal of Geophysical Research, 117(B8), B08104. https://doi.org/10.1029/2012jb009412



- Roberts, A. P., Tauxe, L., Heslop, D., Zhao, X., & Jiang, Z. (2018). A critical appraisal of the "Day" diagram. Journal of Geophysical Research: Solid Earth, 123(4), 2618–2644. https://doi.org/10.1002/2017JB015247
- Robertson, D. J., & France, D. E. (1994). Discrimination of remanence-carrying minerals in mixtures, using isothermal remanent magnetisation acquisition curves. *Physics of the Earth and Planetary Interiors*, 82(3–4), 223–234. https://doi.org/10.1016/0031-9201(94)90074-4
- Sagnotti, L., Macrì, P., Egli, R., & Mondolino, M. (2006). Magnetic properties of atmospheric particulate matter from automatic air sampler stations in Latium (Italy): Toward a definition of magnetic fingerprints for natural and anthropogenic PM₁₀ sources. *Journal of Geophysical Research*, 111(B12), B12S22. https://doi.org/10.1029/2006jb004508
- Song, Y., Guo, Z., Marković, S., Hambach, U., Deng, C., Chang, L., et al. (2018). Magnetic stratigraphy of the Danube loess: A composite titelstari slankamen loess section over the last one million years in vojvodina, Serbia. *Journal of Asian Earth Sciences*, 155, 68–80. https://doi.org/ 10.1016/j.jseaes.2017.11.012
- Spassov, S., Heller, F., Kretzschmar, R., Evans, M. E., Yue, L. P., & Nourgaliev, D. K. (2003). Detrital and pedogenic magnetic mineral phases in the loess/paleosol sequence at Lingtai (Central Chinese Loess Plateau). *Physics of the Earth and Planetary Interiors*, 140(4), 255–275. https:// doi.org/10.1016/j.pepi.2003.09.003
- Stephens, M., Krzyszkowski, D., Ivchenko, A., & Majewski, M. (2003). Palaeoclimate and pedosedimentary reconstruction of a Middle to Late Pleistocene loessepalaeosol sequence. Prymorske, SW Ukraine. *Studia Quaternaria*, *19*, 3e17.
- Stockhausen, H. (1998). Some new aspects for the modelling of isothermal remanent magnetization acquisition curves by cumulative log Gaussian functions. *Geophysical Research Letters*, 25(12), 2217–2220. https://doi.org/10.1029/98GL01580
- van Velzen, A. J., & Dekkers, M. J. (1999). Low-temperature oxidation of magnetite in loess-paleosol sequences: A correction of rock magnetic parameters. Studia Geophysica et Geodaetica, 43(4), 357–375. https://doi.org/10.1023/a:1023278901491
- Zeeden, C., Hambach, U., Veres, D., Fitzsimmons, K., Obreht, I., Bösken, J., & Lehmkuhl, F. (2017). Millennial scale climate oscillations recorded in the Lower Danube loess over the last glacial period. *Palaeogeography, Palaeoclimatology, Palaeoecology, 509*, 164–181. https:// doi.org/10.1016/j.palaeo.2016.12.029
- Zeeden, C., Kels, H., Hambach, U., Schulte, P., Protze, J., Eckmeier, E., et al. (2016). Three climatic cycles recorded in a loess-palaeosol sequence at Semlac (Romania) Implications for dust accumulation in south-eastern Europe. *Quaternary Science Reviews*, 154, 130–154. https://doi.org/10.1016/j.quascirev.2016.11.002
- Zhang, Y., Jia, D., Muxworthy, A. R., Zhang, Y., Li, Y., Yin, H., et al. (2020). Fluid migration and widespread remagnetization in the Dabashan fold and thrust belt, China. *Journal of Geophysical Research: Solid Earth*, *125*(11), e2020JB019989. https://doi.org/10.1029/2020JB019989
- Zhao, X., Fujii, M., Suganuma, Y., Zhao, X., & Jiang, Z. (2018). Applying the Burr type xii distribution to decompose remanent magnetization curves. *Journal of Geophysical Research: Solid Earth*, *123*(10), 8298–8311. https://doi.org/10.1029/2018JB016082
- Zhao, X., Roberts, A. P., Heslop, D., Paterson, G. A., Li, Y., & Li, J. (2017). Magnetic domain state diagnosis using hysteresis reversal curves. Journal of Geophysical Research: Solid Earth, 122(7), 4767–4789. https://doi.org/10.1002/2016JB013683

References From the Supporting Information

- Hyvärinen, A. (1998). New approximations of differential entropy for independent component analysis and projection pursuit. In Advances in Neural Information Processing Systems (Vol. 10, pp. 273–279). MIT Press.
- Miettinen, J., Nordhausen, K., & Taskinen, S. (2018). fICA: FastICA algorithms and their improved variants. *The RUSI Journal*, 10/2(2), 148. https://doi.org/10.32614/RJ-2018-046
- Tichavský, P., Koldovský, Z., & Oja, E. (2007). Speed and accuracy enhancement of linear ICA techniques using rational nonlinear functions. In Proceedings 7th International Conference, ICA 2007, Proceedings, London, 9–12 September (pp. 285–292).